Author's response to reviews

Title: Overlap of Expression Quantitative Trait Loci (eQTL) in Human Brain and Blood

Authors:

Marna Mckenzie (marna.mckenzie@gmail.com)
Anjali K Henders (Anjali.Henders@qimr.edu.au)
Anthony Caracella (Anthony.Caracella@qimrberghofer.edu.au)
Naomi R Wray (naomi.wray@uq.edu.au)
Joseph E Powell (joseph.powell@uq.edu.au)

Version: 3 Date: 7 April 2014

Author's response to reviews:

.